Author's response to reviews

Title: Increasing the proportion of healthier foods available with and without reducing portion sizes and energy purchased in worksite cafeterias: Protocol for a stepped-wedge randomised controlled trial

Authors:

James Reynolds (jpr63@medschl.cam.ac.uk)

Daina Kosīte (kosite.daina@gmail.com)

Brier Rigby Dames (brierrd1@gmail.com)

Laura Brocklebank (Laura.Brocklebank@bristol.ac.uk)

Mark Pilling (mark.pilling@medschl.cam.ac.uk)

Rachel Pechey (rp444@medschl.cam.ac.uk)

Gareth Hollands (gjh44@medschl.cam.ac.uk)

Theresa Marteau (tm388@medschl.cam.ac.uk)

Version: 1 Date: 04 Nov 2019

Author's response to reviews:

All suggested changes have been made.

One other note, the corresponding author during the submission process is listed at James Reynolds. However, upon acceptance the corresponding author to appear on the paper should be Theresa Marteau (as stated in the manuscript)